Author's response to reviews

Title: Measles mimicking HIV seroconversion syndrome: a case report

Authors:

Mahua T Chatterjee (mahua142003@yahoo.co.uk)
Margaret Coleman (megcoleman@doctors.org.uk)
Gary Brook (gary.brook@nwlh.nhs.uk)
Daniel McCrea (daniel.mccrea@nwlh.nhs.uk)

Version: 3 Date: 17 January 2009

Author's response to reviews:

Dear JMCR Editorial Team
Thank you for the reviewers comments which were very helpful. Changes have been made appropriately to the article

Reviewer 2 - Dr Estrada-Y-Martin
appropriately suggested changing the word "mimicking" in the last paragraph of the introduction to "thought to be" - This sentence has been changed to: Here we present an unusual case of measles initially thought to be HIV serocconversion syndrome.

Reviewer 1 - Francois Venter
Has recommended that the article be "published as it is". However he comments that the choice of broad spectrum antibiotics was curious. In the case presentation we have added in the word 'severe' bacterial infection...so the sentence reads - 'another possibility was of an atypical but severe bacterial infection' (hopefully this explains why we used broad spectrum antibiotics - At the time we were faced with a very sick patient and we were covering all options).

The rest of the comments were helpful observations rather than suggestions for change to the article.

We hope that the above changes made as suggested by the reviewers, now make the article acceptable for publication

Yours sincerely Dr Mahua Chatterjee